Author’s response to reviews

Title: Novel variants in a patient with late-onset hyperprolinemia type II: diagnostic key for status epilepticus and lactic acidosis

Authors:

Jeremias Motte (jeremias.motte@rub.de)
Anna Lena Fisse (anna.fisse@rub.de)
Thomas Grüter (thomas.grueter@rub.de)
Ruth Schneider (ruth.schneider@rub.de)
Thomas Breuer (thomas breuer@rub.de)
Thomas Lücke (luecke.thomas@rub.de)
Stefan Krueger (st.krueger@medizinische-genetik-dresden.de)
Huu Phuc Nguyen (Huu.Nguyen-r7w@ruhr-uni-bochum.de)
Ralf Gold (ralf.gold@rub.de)
Ilya Ayzenberg (ilya.ayzenberg@rub.de)
Gisa Ellrichmann (gisa.ellrichmann@rub.de)

Version: 2 Date: 17 Dec 2019

Author’s response to reviews:

Dear Dr. Darren Byrne,
Dear Dr. Madsen,

we are pleased to submit the revised manuscript for publication as a report in the journal BMC Neurology.

We are grateful for your corrections and the significant improvement in the manuscript. A detailed answer to your comments can be found in the attachment.

We hereby declare that there are no authors other than those listed on the title page of this manuscript. We confirm this manuscript has not been accepted for publication elsewhere, is not being considered for publication elsewhere and does not duplicate material already published. We confirm all authors’ have read and approved the submission.

Yours sincerely,
Jeremias Motte, MD
Gisa Ellrichmann, MD (senior author)